CSCI 3291 (Java with Internet Applications)

Homework #0: Background Questionnaire Assigned: August 26, 1999 Due: August 31, 1999 Total points: 5

• Your name:

• Your major and year (e.g., “CS, junior”):

• Will you likely do your homeworks and project for this course on departmental machines, or do you plan to use a home system? If the latter, what?

• Briefly describe your programming background (coursework at Trinity and elsewhere, work experience, etc.).

• Have you used Java before? CC++? Extensively?

• How familiar would you say you are with object-oriented programming?

• What are your objectives in taking this course?